Author's response to reviews

Title: ADAMTS2 gene dysregulation in T/myeloid mixed phenotype acute leukemia

Authors:

Giuseppina Tota (giuseppina.tota@uniba.it)
Nicoletta Coccaro (nicoletta.coccaro@uniba.it)
Antonella Zagaria (antonella.zagaria@uniba.it)
Luisa Anelli (luisa.anelli@uniba.it)
Paola Casieri (paolacasieri@gmail.com)
Angelo Cellamare (cellamareangelo@hotmail.com)
Angela Minervini (minervini.angela@gmail.com)
Crescenzio F Minervini (ezominervini@gmail.com)
Claudia Brunetti (brunetti.claudia@gmail.com)
Luciana Impera (luciana.impera@uniba.it)
Paola Carluccio (paola.carluccio@libero.it)
Cosimo Cumbo (cosimo.cumbo@alice.it)
Giorgina Specchia (giorgina.specchia@uniba.it)
Francesco Albano (francesco.albano@uniba.it)

Version: 2  Date: 22 September 2014

Author's response to reviews: see over
Dear Editor in Chief,

please find enclosed our manuscript entitled “ADAMTS2 gene dysregulation in T/myeloid mixed phenotype acute leukemia” which we are submitting in the hope that it may be found suitable for publication in *BMC Cancer*. We describe a case affected by a rare kind of leukemia (T/myeloid Mixed phenotype acute leukemia) characterized by *ADAMTS2* gene dysregulation which allowed us to monitor the minimal residual disease. The present case underlines the importance of molecular studies in rare leukemias to better understand disease pathogenesis and plan best therapeutic strategies.

According to Editorial request line and page numbers have been included, and Ethics section has been added (Page 8, lines 193-197).

All the authors declare that the material is original research and has not been submitted for publication elsewhere while under consideration. No potential competing interests are declared.

Thanking you in advance for your kind attention,

Francesco Albano, MD, PhD
Hematology, Azienda Ospedaliera Universitaria Policlinico, P.zza G. Cesare, 11 70124 Bari - Italy; Phone +39 (0)80-5478031. Fax +39 (0)80-5508369; e-mail: francesco.albano@uniba.it